1 CS150 Spring 97 Page 1.2. 1

CS150 Newton 1.2. 1

Outline

m Last time:

Ü What is the course all about & why is it important?

Ü What is a digital system? What is a binary digital system?

Ü Boolean Algebra, Truth tables

Ü Operators: inversion, and, or, xor , xnor ( eq )

Ü Design Example: Translating a word problem to a

combinational logic function

Ü Design Example: Translating a word problem into a sequential

design language

Ü State Transition Graph

Ü State Transition Table

Ü Mealy and Moore Forms

m This lecture:

Ü Design versus Implementation

Ü Specification, Description, and Depiction

Ü Conceptual Blocks & How to Overcome Them

Ü Good Versus Bad Design

Ü The Role of Language in Design

Ü The CS150 Project

CS150 Newton 1.2. 2

From Ideas to Manufacturable Products

Conceptualization

Specification Description

Implementation

The process by which one has ideas...

2 CS150 Spring 97 Page 1.2. 2

CS150 Newton 1.2. 3

Specification vs . Description m Specification: Saying what I want; presents behavior in terms of results.

e.g. " A { A[i,j] ¬ 0} m Description: Saying how to do it; describes behavior in terms of procedure or process.

e.g. for(i=0; i<N; i++) for(j=0; j<M; j++)

A[i][j] = 0;

m We do not have specification languages for general-purpose digital design. For some special-purpose applications (e.g. DSP) we do.

CS150 Newton 1.2. 4

But Can We Really Specify or Even Describe

Anything of Real Interest... ?

m It is not clear, in today’s complex world, with complex interactions, that descriptive languages are particularly useful any more:

"